Author’s response to reviews

Title: Stroke knowledge among diabetics: Influence of age, gender, education, and migration status

Authors:

Birgitta M Weltermann (birgitta.weltermann@uk-essen.de)
Yousef Driouach-Bleckmann (docyou_99@yahoo.de)
Sabrina Reinders (sabrina.reinders@uk-essen.de)
Peter Berndt (peter.berndt@uk-essen.de)
Stefan Gesenhues (stefan.gesenhues@uk-essen.de)

Version: 2 Date: 31 May 2013

Author’s response to reviews:

1) Ethics committee: The Ethics Committee of the University Clinic, University Duisburg-Essen, Germany, approved the study. The name is included in the revised manuscript.

2) Authors’ Contributions: they are specified in the revised manuscript.

3) Details about the questionnaire validation and cultural evaluation are now included in the methods section.

Thank you for your support.